Author’s response to reviews

Title: Prenatal Hepatitis B screening and associated factors in a high prevalence district of Lira, Northern Uganda: a community based cross sectional study

Authors:

Paul Katamba (katambacadet@gmail.com)

David Mukunya (zebdaevid@gmail.com)

Doris Kwesiga (dknnkwesiga@gmail.com)

Victoria Nankabirwa (Nankabirwa@gmail.com)

Version: 2 Date: 09 Jul 2019

Author’s response to reviews:

Dear editor,

All the comments were addressed except 2 (overlap). We could not locate the exact statements said to overlap with those in the cited article Because the Line numbering in our manuscript does not match with what you cited. Kindly quote the exact statements as written so that we can act as advised even if we disagree with that allegation.